Author’s response to reviews

Title: Socio-demographic characteristics and co-occurrence of depressive symptoms with chronic diseases among older adults in China: the China Longitudinal Ageing Social Survey

Authors:
Zhenjie Wang (zhenjie.wang@pku.edu.cn)
Hanmo Yang (hmyang2018@nsd.pku.edu.cn)
Zhanyuan Guo (guozhanyuan@pku.edu.cn)
Bei Liu (1801213947@pku.edu.cn)
Shen Geng (2691443787@qq.com)

Version: 5 Date: 20 Sep 2019

Author’s response to reviews:

20 September 2019

Dear Editor,

On behalf of my colleagues, I would like to resubmit a manuscript (ID: BPSY-D-19-00520_R4) entitled “Socio-demographic characteristics and co-occurrence of depressive symptoms with chronic diseases among older adults in China: the China Longitudinal Ageing Social Survey” for publication in BMC Psychiatry. In this submission, we have added the type of consent and provided the reason for choosing consent, which was as editor suggested.

Thank you for considering our manuscript.

Yours sincerely,

Zhenjie Wang on behalf of authors

Institute of Population Research
Peking University
No. 5 Yiheyuan Road, Haidian District
Beijing 100871, People’s Republic of China.